Author’s response to reviews

Title: Searching for chronic hepatitis B patients in a low prevalence area - role of racial origin.

Authors:
Dr Suzane K Ono-Nita (skon@usp.br)
Dr Flair J Carrilho (fjcarril@usp.br)
Rita A Cardoso (rita@statistika.com.br)
Marcelo E Nita (marcelo_nita@uol.com.br)
Luiz C Da Silva (lucadasi@bol.com.br)

Version: 5 Date: 12 Mar 2004

Cover letter

Re: Searching for chronic hepatitis B patients in a low prevalence area - role of racial origin.
1180107372542836

Dear Dr. Christabel,

Thank you very much for your e-mail and for considering our manuscript for publication. The revisions of the referees were very helpful and we believed that it improved our manuscript considerably.

We changed the manuscript as suggested by Dr. Prieto as follow:

1. the term "prevalence" were substituted by "frequency".
2. On the methods section, reference to ALT levels in relatives was omitted.
3. In the abstract section, the term "trial" was replaced by "study". Also, where it reads "..were screened for the HBV serological markers" it was changed to "..were screened for HBV serological markers".
4. On page 4, second paragraph, where it reads "comprises clinical..." it was changed to "comprised clinical..".

We would like to thank in advance for your kind consideration,

Regards,

Suzane Kioko Ono-Nita